Author's response to reviews

Title: Difference in overall and age-specific prevalence of human papillomavirus infection in Italy: evidence from NTCC trial

Authors:

Iacopo Baussano (baussano@iarc.fr)
Silvia Franceschi (franceschis@iarc.fr)
Anna Gillio-Tos (gilliotos.demarco@cpo.it)
Francesca Carozzi (f.carozzi@ispo.toscana.it)
Massimo Confortini (m.confortini@ispo.toscana.it)
Paolo Dalla Palma (paolo.dallapalma@apss.tn.it)
Margherita De Lillo (anapat@ausl.imola.bo.it)
Annarosa Del Mistro (annarosa.delmistro@ioveneto.it)
Laura De Marco (gilliotos.demarco@cpo.it)
Carlo Naldoni (cnaldoni@regione.emilia-romagna.it)
Paola Pierotti (paola.pierotti@ausl.bologna.it)
Patrizia Schincaglia (p.schincaglia@ausl.ra.it)
Nereo Segnan (nereo.segnan@cpo.it)
Manuel Zorzi (manuel.zorzi@ioveneto.it)
Paolo Giorgi-Rossi (giorgirossi@asplazio.it)
Guglielmo Ronco (guglielmo.ronco@cpo.it)

Version: 2 Date: 25 April 2012

Author's response to reviews: see over
Dear Dr. Manibo,

We thank you for your e-mail of 18 April 2012 regarding our manuscript entitled "Difference in overall and age-specific prevalence of human papillomavirus infection in Italy: evidence from NTCC trial" by Baussano et al.

As per your request, we have made the necessary modifications in the Methods section, page 5, second paragraph. Modifications are highlighted in yellow for ease of reference.

We very much hope BMC Infectious Diseases will consider our manuscript suitable for publication, and look forward to your response.

Yours sincerely,

Dr Iacopo Baussano
Scientist, Infections and Cancer Epidemiology Group
baussano@iarc.fr

ENCL: (1)